Author's response to reviews

Title: Homeopathic treatment of patients with chronic sinusitis: A prospective observational study with 8 years follow-up

Authors:

Claudia M Witt (terw@gmx.de)
Rainer Lüdtke (r.luedtke@carstens-stiftung.de)
Stefan N Willich (stefan.willich@charite.de)

Version: 3 Date: 17 July 2009

Author's response to reviews: see over
BMC Ear, Nose and Throat Disorders

The Editor

Formatting Changes "Homeopathic Treatment of Patients with Sinusitis – A Prospective Observational Study with 8 Years Follow-Up"

Dear Editor,

We would like to submit the above manuscript incorporating the desired changes.

These were:

- The Abstract section headings were renamed as requested.
- A sentence stating that all authors read and approved the final manuscript was added to Authors' contributions.
- References 1, 15, 33 were corrected.
- Tables were reformatted to fit into an A4 portrait page.
- Title and authors' names were set to sentence case.
- Authors' qualifications were removed.
- Keywords, word-counts, running head and header were removed from the title page. Abbreviations were formatted as a paragraph.
- Figure placement text was removed.

On behalf of Prof. Dr. Witt,

Yours sincerely,

Thorolf Weißhuhn